Dear Editors:

We have made the formatting changes as specified in your email of 11/7/05, correcting capitalization, endnote formats, removing the keyword section, moving funding to acknowledgements, removing word count, revising formatting for references, adding sections on competing interests and on authors’ contributions, moving tables and appendices are requested and removing the second heading from the last table. Thank you for the opportunity to revise the manuscript.